Author's response to reviews

Title: Portal, Superior Mesenteric and Splenic Vein Thrombosis Secondary to Hyperhomocysteinemia with Pernicious Anemia: A Case Report

Authors:

Prashanth Venkatesh (prashanth_1405@yahoo.com)
Nissar Shaikh (nissatfirdous99@gmail.com)
Mohammad F Malmstrom (mmalmstrom@hmc.org.qa)
Vajjala R Kumar (vkumar@hmc.org.qa)
Bakr Nour (bmn2001@qatar-med.cornell.edu)

Version: 4 Date: 20 June 2014

Author's response to reviews:

Dear Sir,

We have made the following changes to the most recently submitted version of the manuscript as per your wishes:

1. Title page added with complete list of author names, affiliations and Email addresses
2. Cover sheet removed from manuscript
3. Keywords added
4. Ethnicity of patient included in both abstract and case presentation sections
5. Highlighted portions in previous version modified
6. List of abbreviations added
7. Consent section added
8. Competing interests section added
9. Authors’ Contributions section added
10. Acknowledgments section added
11. Figure legend section added

Thanking you,
Yours sincerely,
Prashanth Venkatesh, MD